Author's response to reviews

Title: High prevalence of lipoatrophy in pre-pubertal South African children on antiretroviral therapy: A cross-sectional study

Authors:

Steve Innes (innes@sun.ac.za)
Mark F Cotton (mcot@sun.ac.za)
Richard Haubrich (rhaubrich@mail.ucsd.edu)
Maria M Conrading (rc2@sun.ac.za)
Margaret van Niekerk (mbunting@sun.ac.za)
Clair Edson (claedson@pgwc.gov.za)
Helena Rabie (hrabie@sun.ac.za)
Sonia Jain (sojain@ucsd.edu)
Xiaoying Sun (xisun@ucsd.edu)
Ekkehard Zöllner (zollner@sun.ac.za)
Stephen Hough (fsh@sun.ac.za)
Sara H Browne (shbrowne@ucsd.edu)

Version: 2 Date: 16 April 2012

Author's response to reviews: see over
Dear Tom

RE: High prevalence of lipoatrophy in pre-pubertal South African children on antiretroviral therapy: A cross-sectional study

As requested, the following changes have been made:

- Permission to publish the photo’s has been obtained. I have included a statement to that effect in the legend of the photo’s.
- I have added an authors’ contribution statement before the acknowledgments.

Best regards

Steve Innes
Children’s Infectious Diseases Clinical Research Unit (KID-CRU)
Tygerberg Children’s Hospital and Stellenbosch University, Cape Town, South Africa
Tel: +27 21 938 4297 / Fax: +27 21 938 4151 / Email: innes@sun.ac.za